Author's response to reviews

Title: Evaluation of tuberculosis infection control measures implemented at primary health care facilities in Kwazulu-Natal province of South Africa

Authors:

Ntambwe Malangu (gustaymalangu@gmail.com)
Michah Mngomezulu (Michah_Mngomezulu@embanet.com)

Version: 5 Date: 22 November 2014

Author's response to reviews: see over
Dear Editor

RE: Ref. No.: 774- Knowledge and practices about Multi-Drug Resistant Tuberculosis among health care workers in Maseru

<table>
<thead>
<tr>
<th>Issues Raised by Reviewers</th>
<th>Authors’ Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>Suggestions on using one decimal point for most figures such as age, sex, % health workers etc.</td>
<td>All percentages have been set to one decimal throughout the whole paper from the Abstract to the Discussion. This is now highlighted in RED</td>
</tr>
<tr>
<td>Tables will have some administrative edits</td>
<td>The all 4 tables have been resized and reformatted with font size 10. The heading of the table is made of two merged rows to contain the wrapped text</td>
</tr>
</tbody>
</table>

Best regards

Dr. N. Malangu

November 23, 2014